

Clinical Practice Guideline Monitoring children and young people with, or at risk of developing Autosomal Dominant Polycystic Kidney Disease (ADPKD)

Final Version: November 2018 Review Date: November 2023

Authors:

Dr Jan Dudley - Chair

Consultant Paediatric Nephrologist, Bristol

Professor Paul Winyard - Clinical Lead

Consultant Paediatric Nephrologist, London

Dr Matko Marlais - Clinical Lead

Sub-specialty Trainee in Paediatric Nephrology, London

Dr Oliver Cuthell

Locum Consultant Paediatrician, Plymouth

Mrs Tess Harris

CEO Polycystic Kidney Disease Charity (lay member)

Dr Jiehan Chong

Academic Clinical Fellow (Nephrology), Sheffield

Professor John Sayer

Professor of Renal Medicine, Newcastle

Dr Daniel P Gale

Consultant Nephrologist, London

Mrs Lucy Moore

Patient Representative

Mrs Kay Turner

Patient Representative

Ms Sarah Burrows

Community nurse specialist, renal genetics, Birmingham

Dr Richard Sandford

Regional Specialty Advisor in Medical Genetics, Cambridge



Delphi Panelists:

Professor Albert Ong, Professor of Renal Medicine, Sheffield
Professor Neil Turner, Professor of Renal Medicine, Edinburgh
Dr Catriona Shaw, Consultant Nephrologist, London
Dr Yincent Tse, Consultant Paediatric Nephrologist, Newcastle
Dr Heather Lambert, Consultant Paediatric Nephrologist, Newcastle
Dr Rodney Gilbert, Consultant Paediatric Nephrologist, Southampton
Dr Charles Shaw-Smith, Consultant in Clinical Genetics, Exeter
Dr Laura Yates, Consultant in Clinical Genetics, Newcastle
Dr Christine Patch, Consultant Genetic Counsellor, London
Mrs Melanie Azari, lay panellist
Mr Stuart Blount, lay panellist
Mrs Jane Whitney, lay panellist
Dr Sarah Harding, General Practitioner, Leeds

Dr Rebecca Mann, Consultant Paediatrician with Nephrology interest, Taunton
Dr Adamu Sambo, Consultant Paediatrician with Nephrology interest, Gloucester
Dr Mona Aslam, Consultant Paediatrician with Nephrology interest, North West Anglia

Dr Richard Pratt, General Practitioner, Truro

Endorsements



The National Institute for Health and Care Excellence (NICE) has accredited the process used by the Renal Association to produce its Clinical Practice Guidelines. Accreditation is valid for 5 years from January 2017. More information on accreditation can be viewed at www.nice.org.uk/accreditation

Conflicts of Interest Statement

All authors made declarations of interest in line with the policy in the Renal Association Clinical Practice Guidelines Development Manual. Further details can be obtained on request from the Renal Association.

Acknowledgements

This document has been externally reviewed by key stake holders according to the process described in the Clinical Practice Guidelines Development Policy Manual.



Method used to arrive at a recommendation

Evidence reviews were undertaken that focused on literature in relation to children and young people (CYP) with, or are at risk of developing Autosomal Dominant Polycystic Kidney Disease (ADPKD). Medline (1980 – December 2017), EMBASE (1980 – December 2017) and PsychInfo databases were searched as well as websites of national associations in this field. A search strategy was developed by the guideline committee to ensure that all papers addressing the questions were identified using search terms based on PICO methodology (table 1). The clinical leads also hand searched reference lists of reviews and included papers.

Abstracts were screened for relevance by a clinical lead and 1 other member of the guideline committee according to pre-defined inclusion and exclusion criteria as detailed in the scope. Abstracts identified for review by the two reviewers were compared and any disputed abstracts were resolved by the guideline committee. The full papers were then reassessed by the clinical lead to further exclude any study that does not meet the following predefined criteria: Randomised controlled trials (RCT), non-randomised studies if adjusted for key confounders (age, health at baseline, co-morbidities).

Clinicians on the guideline committee critically appraised any eligible papers using critical appraisal skills programme tools. ⁽¹⁾ Where evidence was lacking, formal Delphi consensus methodology was employed. A Delphi panel was constituted, comprising representation from each specialist area covered by the guideline: Nephrology services (3 adult and 3 paediatric nephrologists), clinical genetics (3 representatives), paediatrics with an interest in nephrology (3 representatives), lay members (3) and general practitioners (3 invited, 2 responded). A Likert scale was used for panellists to provide their responses to statements. Consensus agreement and disagreement was defined as 80% of panellists selecting 'agree' or 'disagree' respectively. Individual responses were anonymised to panellists and the working group, with the exception of the chair. No literature was sent to participants to avoid risk of bias. The process was iterative (participants able to change their views in subsequent rounds). Three rounds were undertaken.

Table 1: PICO characteristics

Population	Intervention	Comparison	Outcome	Study design
	Ultrasonography		Mortality	
Children (<18)		Any intervention		Randomised
with a confirmed	Cranial imaging	compared with	Hospitalisations	controlled trials (RCT)
diagnosis of		any other or no		
ADPKD or at risk	Blood Pressure	intervention	Chronic Kidney	Non-randomised
of ADPKD due to	monitoring		Disease	studies if adjusted for
their family				key confounders (age,
history	Monitoring for		Cardiovascular	health at baseline, co-
	albuminuria		disease	morbidities).
	Therapeutics		Hypertension	
	Genetic counselling			



Table of Contents

1.	Introduction	5
2	Summary of clinicial practice guidelines	6
3.	Summary of audit measures	7
1	Summary of research recommendations	0
4.	Summary of research recommendations	8
5.	Rationale for clinical practice guidelines	9
6.	Rationale for research recommendations	16
_		
7.	Lay summary	18



1. Introduction

This guideline makes recommendations for monitoring children and young people (CYP) up to 18 years of age with, or at risk of developing Autosomal Dominant Polycystic Kidney Disease (ADPKD).

ADPKD is the commonest inherited renal disease with an incidence of around 1 in 1000 and accounts for 5-7% of adults commencing renal replacement therapy ^(1.2). Whilst ADPKD has traditionally been thought of as an adult disease, with established renal failure tending to occur in or after the 6th decade, there is clear evidence of earlier manifestation in children and young people (CYP), in whom hypertension and proteinuria may accelerate progression to later stages of chronic kidney disease. There is wide variation in clinical practice facing CYP with confirmed or a family history of ADPKD, with regard to a) assessment of blood pressure and urine testing for the presence of proteinuria b) ultrasound testing to evaluate presence of cysts and c) genetic counselling and testing. In order to improve quality of care and reduce variation in practice, the British Association for Paediatric Nephrology (BAPN) and the UK Renal Association (RA) in collaboration with key partners, has undertaken this work to develop best practice guidance in the area.

- 1. Igarashi P, Somlo S. Genetics and pathogenesis of polycystic kidney disease. J Am Soc Nephrol. 2002 Sep; 13(9):2384-98.
- 2. Shaw C, Picher D, Sandford R et al. Demographics and outcomes study in patients with autosomal dominant polycystic kidney disease (ADPKD) and end stage renal failure (ERF): A UK renal registry analysis on behalf of the ADPKD study group. UK Renal registry annual report 2013



2. Summary of clinical practice guidelines

Guideline 1

We recommend that parents or carers of children at risk of developing ADPKD should be offered information on ADPKD inheritance and potential benefits and harms of testing for ADPKD, by health professionals with specialist knowledge in this area. (1D)

Guideline 2

We recommend that children and young people aged 5 years and above with, or at risk of developing ADPKD, should have an assessment of blood pressure (BP) at least once every 2 years. (1B)

Guideline 3

We recommend that the decision to test for ADPKD in asymptomatic children and young people (CYP) at risk of developing ADPKD, should be undertaken jointly between health professionals and parents or carers and, wherever possible, the young person. (1D)

Guideline 4

If testing is decided on, we suggest that either kidney ultrasound or genetic testing may be offered to asymptomatic children and young people at risk of ADPKD, where testing has been agreed by parents or carers (and, wherever possible, the young person) and health professionals (2D)

Guideline 5

We suggest that, if asymptomatic children at risk of ADPKD do not have cysts on ultrasound, further ultrasound testing should be deferred until adolescence (15-18 years), or later if preferred by the young person (2D)

Guideline 6

We recommend that if genetic testing is planned in children and young people at risk of ADPKD, identification of the mutation in the affected adult family member (if not already known) should be undertaken prior to testing the child or young person. (1D)



3. Summary of audit measures

- Audit Measure 1: Proportion of parents or carers of children at risk of developing ADPKD offered information on ADPKD inheritance and potential benefits and harms of testing for ADPKD
- Audit Measure 2: Proportion of children and young people aged 5 years and above with, or at risk of developing ADPKD, having an assessment of blood pressure (BP) at least once every 2 years
- Audit Measure 3: Proportion of asymptomatic children and young people at risk of developing ADPKD offered testing for ADPKD

• Audit Measure 4:

- **a.** Proportion of asymptomatic children and young people at risk of developing ADPKD offered genetic testing for ADPKD
- **b.** Proportion of asymptomatic children and young people at risk of developing ADPKD offered ultrasound testing for ADPKD
- Audit Measure 5: Proportion of asymptomatic children at risk of ADPKD who do not have cysts on ultrasound, having repeated ultrasound testing prior to adolescence (15-18 years)
- Audit Measure 6: Proportion of asymptomatic children at risk of ADPKD whose parents have been tested for a genetic mutation prior to the child being tested



4. Summary of research recommendations

Research recommendation 1: In children and young people with ADPKD, does regular (e.g. yearly or every 2 years) urine albumin: creatinine monitoring and treatment reduce disease progression?

Research recommendation 2: In children and young people with ADPKD what is a) the incidence of subarachnoid haemorrhage and b) the prevalence of intracranial aneurysm?

Research recommendation 3: In adults, children and young people with ADPKD with a family history of intracranial aneurysm or sub-arachnoid haemorrhage does Intracranial Magnetic Resonance (MR) imaging reduce the risk of intracranial events?



5. Rationale for Clinical Practice Guidelines

Guideline 1

We recommend that parents or carers of children and young people at risk of developing ADPKD should be offered information on ADPKD inheritance and potential benefits and harms of testing for ADPKD, by health professionals with specialist knowledge in this area. (1D)

• Audit Measure 1: Proportion of parents or carers of children at risk of developing ADPKD offered information on ADPKD inheritance and potential benefits and harms of testing for ADPKD

Rationale

No relevant studies were identified for this review question; however, NICE guidance on patient experience in adult NHS services recommends that patients should be provided with information, and the support they need to promote their active participation in care and self-management. This should include information about relevant treatment options and services that they are entitled to, even if these are not provided locally¹. There was 100% agreement with this recommendation in the Delphi consensus process. Health professionals should be aware of the anxiety suffered by families relating to uncertainty of diagnosis as well as at times of testing for ADPKD.

Health professionals should discuss the limitations of testing modalities for ADPKD. Renal ultrasound cannot effectively exclude ADPKD in children. In a retrospective cohort study of ultrasound assessment in children and young people under the age of 15 at risk of ADPKD, 193/420 CYP were diagnosed with cysts at baseline visit (mean age 8.6 + /- 4.2 years) with 227 having no cysts (8.0 ± 4.1 years). In follow up to age fifteen, 18/77 (23%) of the latter group who underwent repeat ultrasound developed cysts. In other words, 23% of CYP with no cysts visible on initial renal ultrasound who received a further ultrasound, were diagnosed with ADPKD (2). No standardised criteria for renal ultrasound diagnosis of ADPKD exist for children under the age of 15 years, and even below the age of 30, there is a significant false negative rate. This is particularly so in families carrying a *PKD2* mutation, which is typically associated with milder disease. In these families, 16.5% of patients between the ages of 15 - 29, with no cysts on initial ultrasound scan, go on to a diagnosis of ADPKD later in life⁴. While genetic testing is more definitive, it is most likely to be informative in families of known genotype, as around 10% of families phenotypically affected by ADPKD do not carry a pathogenic mutation of *PKD1* or *PKD2* that is detected by current technologies⁵. Health professionals (generally nephrologists, geneticists or paediatricians with interest in nephrology) providing information on testing should have a good understanding of these issues.

- 1. National Institute for Health and Care Excellence. Patient experience in adult NHS services: improving the experience of care for people using adult NHS services. NICE guideline (CG138). (2012).
- Gabow, P. A., Kimberling, W. J., Strain, J. D., Manco-Johnson, M. L. & Johnson, A. M. Utility of ultrasonography in the diagnosis of autosomal dominant polycystic kidney disease in children. *J. Am. Soc.* Nephrol. 8, 105–110 (1997).



- **3.** Reed, B. Nobakht E, Dadgar S, et al. Renal Ultrasonographic Evaluation in Children at Risk of Autosomal Dominant Polycystic Kidney Disease. *Am. J. Kidney Dis.* **56,** 50–56 (2010).
- **4.** Pei, Y. *et al.* Unified Criteria for Ultrasonographic Diagnosis of ADPKD. *J. Am. Soc. Nephrol.* **20,** 205–212 (2009).
- 5. Audrézet, M.-P. *et al.* Autosomal dominant polycystic kidney disease: Comprehensive mutation analysis of PKD1 and PKD2 in 700 unrelated patients. *Hum. Mutat.* **33**, 1239–1250 (2012).

Guideline 2

We recommend that children and young people aged 5 years and above with, or at risk of developing ADPKD, should have an assessment of blood pressure (BP) at least once every 2 years. (1B)

Audit Measure 2: Proportion of children and young people aged 5 years and above with, or at risk of developing ADPKD, having an assessment of blood pressure (BP) at least once every 2 years

Rationale

Low quality evidence identified for this review question was deemed by the committee to be sufficient to make a recommendation without the need for formal consensus. A recent systematic review of 928 children with ADPKD across 14 studies estimated the prevalence of hypertension to be 20% (95% CI 15-27%)⁽¹⁾. There is also evidence to suggest that children with ADPKD whose BP is high or borderline high have an increased left ventricular mass index on echocardiography, a marker of cardiac target organ damage⁽²⁾. Studies have also shown that children with ADPKD who are hypertensive show an increased total kidney volume, compared to those that are normotensive⁽³⁻⁵⁾. There are, as yet, no studies assessing long term outcomes for the treatment of hypertension in children with ADPKD. However, there is clear evidence of benefit of treating hypertension in children generally and especially in those with chronic kidney disease⁽⁶⁾.

We recommend that both children with a confirmed diagnosis of ADPKD and those at risk of ADPKD through family history should have BP monitored, since some families may choose not to undertake testing for ADPKD in childhood. The risk of developing hypertension in children with ADPKD rises with age⁽¹⁾. Hypertension under the age of 5 years is uncommon in ADPKD⁽⁷⁾ and should prompt a search for an alternative diagnosis. In the absence of any other disease process BP rises slowly in children with ADPKD, therefore monitoring BP every 2 years should be sufficient to detect a rise in BP requiring treatment during childhood.

- 1. Marlais et al. Hypertension in autosomal dominant polycystic kidney disease: a meta-analysis. Arch Dis Child 2016, 101 (12): 1142-1147.
- 2. Cadnapaphornchai et al. Increased left ventricular mass in children with autosomal polycystic kidney disease and borderline hypertension. Kidney Int 2008, 74: 1192-1196.



- 3. Fick-Brosnahan et al. Progression of autosomal-dominant polycystic kidney disease in children. Kidney Int 2001, 59: 1654-1662.
- **4.** Seeman et al. Ambulatory blood pressure correlates with renal volume and number of renal cysts in children with autosomal dominant polycystic kidney disease. Blood Press Monit 2003, 8:107-110.
- **5.** Cadnapaphornchai et al. Magnetic resonance imaging of kidney and cyst volume in children with ADPKD. Clin J Am Soc Nephrol 2011, 6: 369-376.
- **6.** The ESCAPE Trial Group. Strict Blood-Pressure Control and Progression of Renal Failure in Children. N Engl J Med 2009, 361: 1639-1650.
- 7. Boyer et al. Prognosis of autosomal dominant polycystic kidney disease diagnosed in utero or at birth. Pediatr Nephrol 2007,22: 380-388.

Guideline 3

We recommend that the decision to test for ADPKD in asymptomatic children and young people (CYP) at risk of developing ADPKD, should be undertaken jointly between health professionals and parents or carers and, wherever possible, the young person. (1D)

Audit Measure 3: Proportion of asymptomatic children and young people at risk of developing ADPKD offered testing for ADPKD

Rationale

No relevant studies were identified for this review question. There was 88% agreement with this recommendation in the Delphi consensus process.

Inheriting and passing on a genetic disease may have significant psychological impact, ranging from frank depression through anxiety, guilt, anger, uncertainty and sadness ⁽¹⁾. This genetic anxiety or 'guilt' may increase with uncertainty over disease variability and where there is a perceived lack of diagnosis or effective therapy. These are common issues in CYP with ADPKD, which may be further compounded by mixed messages from the medical community, with conflicting opinions as to the clinical significance of ADPKD in childhood. Genetic Counselling and testing for a condition in at risk individuals has been reported to ameliorate the psychological burden for some families and children⁽²⁻⁴⁾, however, wherever possible, the child or young person should be involved in this decision. Article 12 of the United Nations Convention on the Rights of the Child (1989) states that we must assure 'the child who is capable of forming his or her own views has the right to express those views freely' and that 'the views are given due weight in accordance with the age and maturity of the child' ⁽⁵⁾.



Important issues to be discussed in counselling before deciding to test for ADPKD include:

- Age and 'Gillick' competence of the CYP and consideration of whether to wait until they can make a fullyinformed decision.
- Risk of false negative results for ultrasound and genetic tests as described on page 9
- Implications for ongoing management. Blood pressure monitoring is likely to be as important as making the
 diagnosis in preventing long term complications, but this should be balanced against the psychological
 benefits of confirming / refuting (noting risk of false negative results) ADPKD. The discussion is likely to
 change when new treatments become available for children and young people with ADPKD.

A recent European ADPKD Forum multidisciplinary position statement states that 'Individuals with ADPKD should have access to lifelong, multidisciplinary, specialist and patient-centred care, with information and support to help patients and their families act as fully informed and active partners in care ⁽⁶⁾.

References

- 1. McAllister M, Davies L, Payne K et al. The emotional effects of genetic diseases: implications for clinical genetics. Am J Med Genet A. 2007 Nov 15;143A(22):2651-61.
- 2. Metcalfe A, Plumridge G, Coad J et al. Parents' and children's communication about genetic risk: a qualitative study, learning from families' experiences. *Eur J Hum Genet*. 2011;19(6):640-646.
- 3. Rowland E, Metcalfe A. Communicating inherited genetic risk between parent and child: a meta-thematic synthesis. *Int J Nurs Stud.* 2013;50(6):870-880.
- **4.** Patch, C, Middleton, A. (2018). Genetic counselling in the era of genomic medicine. *British Medical Bulletin*, 2018; 126 (1) 27–36
- **5.** United Nations Convention on the Rights of the Child (1989) http://www.un.org/documents/ga/res/44/a44r025.htm
- 6. Harris T, Sandford R, de Coninck B et al. European ADPKD Forum multidisciplinary position statement on autosomal dominant polycystic kidney disease care: European ADPKD Forum and Multispecialist Roundtable participants. Nephrol Dial Transplant. 2017 Dec 22. doi: 10.1093/ndt/gfx327. [Epub ahead of print] PMID: 29309655

Guideline 4

If testing is decided on, we suggest that either kidney ultrasound or genetic testing may be offered to asymptomatic children and young people at risk of ADPKD, where testing has been agreed by parents or carers (and, wherever possible, the young person) and health professionals (2D)



Audit Measure 4:

- **a.** Proportion of asymptomatic children and young people at risk of developing ADPKD offered genetic testing for ADPKD
- **b.** Proportion of asymptomatic children and young people at risk of developing ADPKD offered ultrasound testing for ADPKD

Rationale

No relevant studies comparing outcomes in CYP at risk of ADPKD undergoing genetic versus ultrasound testing for ADPKD were identified for this review question. There was 70% agreement with this recommendation (statement) by the Delphi panellists, i.e. consensus not reached. Areas for disagreement largely related to concerns about the utility of ultrasound, particularly in younger children with no standardised criteria for renal ultrasound diagnosis of ADPKD under the age of 15 years and significant false negative rates ^(1,2). The importance of involving CYP in the decision making process was also noted in the Delphi responses. The committee believed that to offer genetic but not ultrasound testing to asymptomatic children and young people at risk of ADPKD could result in a significant change in practice in the absence of evidence and that the importance of individual choice between the two testing modalities should be emphasised given the lack of clear superiority of one over the other. The committee also noted that, although not reaching consensus, the majority (70%) of panellists supported offering a choice of kidney ultrasound or genetic testing to asymptomatic CYP at risk of ADPKD, where testing has been agreed by parents or carers and health professionals (and, wherever possible, the young person) and agreed to make a weaker 'we suggest' recommendation.

References

- **1.** Gabow, P. A., Kimberling, W. J., Strain, J. D., Manco-Johnson, M. L. & Johnson, A. M. Utility of ultrasonography in the diagnosis of autosomal dominant polycystic kidney disease in children. *J. Am. Soc. Nephrol.* **8**, 105–110 (1997).
- 2. Reed, B. *et al.* Renal Ultrasonographic Evaluation in Children at Risk of Autosomal Dominant Polycystic Kidney Disease. *Am. J. Kidney Dis.* 56, 50–56 (2010).

Guideline 5

We suggest that, if asymptomatic children at risk of ADPKD do not have cysts on ultrasound, further ultrasound testing should be deferred until adolescence (15-18 years), or later if preferred by the young person (2D)

Audit Measure 5: Proportion of asymptomatic children at risk of ADPKD who do not have cysts on ultrasound, having repeated ultrasound testing prior to adolescence (15-18 years)



Rationale

No relevant studies assessing outcomes in asymptomatic children at risk of ADPKD undergoing repeated ultrasound testing for ADPKD, compared with those not undergoing repeated ultrasound testing, were identified for this review question. There was 70% agreement with this recommendation (statement) by the Delphi panellists, i.e. consensus not reached. Areas for disagreement largely related to the age of the child at the time of the first negative ultrasound scan and a preference to perform more regular ultrasound scans if requested by parents. There may also be concern by health professionals about the risk of missing very early onset ADPKD, however, these children comprise less than 1% of cases and are likely to be genetically distinct from typical ADPKD, perhaps with multiple compound mutations combining to generate such an aggressive phenotype ^(1,2). This recommendation does not apply to these children, who undergo diagnostic testing for symptomatic disease (e.g. severe renal enlargement, hypertension) and are found to have radiological abnormalities at presentation.

Repeated ultrasound testing is not routinely undertaken in adults with ADPKD where significant progression usually occurs over decades. In the absence of evidence of benefit of repeated ultrasound testing in asymptomatic CYP, together with its resource implications, and the importance of considering the wishes of the young person in the decision making process where possible, the committee was of the view that a recommendation for repeated ultrasound testing prior to adolescence could not be made. The committee considered whether repeated ultrasound testing might help to alleviate parental anxiety, but noted in their experience that repeated ultrasound testing may also add to the psychological burden for families.

The committee also noted that, although not reaching consensus, the majority (70%) of panellists supported deferring further ultrasound testing until adolescence (15-18 years, or later if preferred by the young person) in asymptomatic children at risk of ADPKD who do not have cysts on initial ultrasound. The committee therefore agreed to make a weaker 'we suggest' recommendation.

References

- 1. Bergmann C, et al. Mutations in multiple PKD genes may explain early and severe polycystic kidney disease. J Am Soc Nephrol. 2011;22(11):2047–2056
- 2. Rossetti S1, Kubly VJ, Consugar MB, Hopp K, Roy S, Horsley SW, Chauveau D, Rees L, Barratt TM, van't Hoff WG, Niaudet P, Torres VE, Harris PC. Incompletely penetrant PKD1 alleles suggest a role for gene dosage in cyst initiation in polycystic kidney disease. Kidney Int. 2010 Feb;77(4):368.

Guideline 6

We recommend that if genetic testing is planned in children and young people at risk of ADPKD, identification of the mutation in the affected adult family member (if not already known) should be undertaken prior to testing the child or young person. (1D)

Audit Measure 6: Proportion of asymptomatic children at risk of ADPKD whose parents have been tested for a genetic mutation prior to the child being tested



Rationale

No relevant studies were identified for this review question. There was 88% agreement with this recommendation in the Delphi consensus process. ADPKD is associated with a wide range of mutations in the genes PKD1 and PKD2, many of which occur only in one family¹. As a result, genetic testing of PKD1 and PKD2 can find sequence variations that have not been previously described, and it can be difficult to determine the pathogenicity of these changes. First identifying the mutation in the affected adult family member allows for the use of segregation analysis to help assign pathogenicity to any mutations identified in the child or young person. Furthermore, up to 10% of ADPKD is not associated with detectable mutations in PKD1 or PKD2². This means that failure to identify a PKD1 or PKD2 mutation in a predictive genetic test cannot reliably exclude ADPKD, unless the familial mutation is known be in those genes.

- 1. 1. PKD Mutation Database. (2016). Available at: http://pkdb.mayo.edu/cgi-bin/v2_display_mutations.cgi?apkd_mode=PROD. (Accessed: 22nd April 2016)
- 2. Audrézet, M.-P. *et al.* Autosomal dominant polycystic kidney disease: Comprehensive mutation analysis of PKD1 and PKD2 in 700 unrelated patients. *Hum. Mutat.* **33**, 1239–1250 (2012).



6. Rationale for research recommendations

Research recommendation 1: In children and young people with ADPKD does regular (e.g. yearly or every 2 years) urine albumin: creatinine monitoring and treatment improve outcome?

Rationale

The guideline committee was not able to make a recommendation on monitoring of urine albumin: creatinine in CYP with ADPKD or at risk of ADPKD. No relevant studies assessing outcomes in CYP with proteinuria were identified and no consensus was achieved in 2 rounds of the Delphi survey; only 41% of panellists agreed with the statement 'Urine protein estimation (best assessed at urine albumin:creatinine) should be offered at least every 2 years to children and young people with confirmed ADPKD commencing at 5 years of age'. The prevalence of proteinuria in CYP with ADPKD has been reported to be as high as 20% ⁽¹⁾, however, a number of studies have failed to show a relationship between hypertension and proteinuria in children with ADPKD ⁽¹⁻³⁾. In adults with ADPKD, established proteinuria and microalbuminuria are reported to be associated with increased mean arterial pressure and more severe renal cystic involvement ^(4,5).

References

- 1. Marlais et al. Hypertension in autosomal dominant polycystic kidney disease: a meta-analysis. Arch Dis Child 2016, 101 (12): 1142-1147
- 2. Fick-Brosnahan et al. Progression of autosomal-dominant polycystic kidney disease in children. Kidney Int 2001, 59: 1654-1662
- **3.** Selistre et al. Early renal abnormalities in children with postnatally diagnosed autosomal dominant polycystic kidney disease. Pediatric Nephrology 2012, 27: 1589-1593
- 4. Chapman AB, Guay-Woodford LM, Grantham JJ. Renal structure in early autosomal-dominant polycystic kidney disease (ADPKD): The Consortium for Radiologic Imaging Studies of Polycystic Kidney Disease (CRISP) cohort. Kidney International 2003; 64: 1035–1045
- **5.** Chapman AB, Johnson AM, Gabow PA. Overt proteinuria and microalbuminuria in autosomal dominant polycystic kidney disease. J Am Soc Nephrol. 1994;5(6):1349-54.

Research recommendation 2: In children and young people with ADPKD what is a) the incidence of subarachnoid haemorrhage and b) the prevalence of intracranial aneurysm?

Research recommendation 3: In adults, children and young people with ADPKD with a family history of intracranial aneurysm or sub-arachnoid haemorrhage does Intracranial Magnetic Resonance (MR) imaging reduce the risk of intracranial events?



Rationale

The guideline committee was not able to make a recommendation on MR imaging in CYP with ADPKD or at risk of ADPKD. No relevant studies examining whether neurological imaging in CYP with ADPKD and a family history of intracranial events is associated with reduced cardiovascular morbidity compared with those who do not have neurological imaging, were identified. Three individual case reports of people under 18 years with ADPKD with intracranial events were found but did not meet inclusion criteria.

No consensus was achieved in 2 rounds of the Delphi survey; 61% of panellists agreed with the statement 'Intracranial Magnetic Resonance (MR) imaging should be offered to CYP with ADPKD with a family history of intracranial aneurysm or sub-arachnoid haemorrhage' (round 1), whilst 41% agreed with the statement 'Intracranial Magnetic Resonance (MR) imaging should not routinely be offered to CYP with ADPKD even with a family history of intracranial aneurysm or sub-arachnoid haemorrhage' (round 2). The lack of published data around screening in CYP<18y was acknowledged and concern was noted with regard to thresholds for neurosurgical intervention if intracranial aneurysms were identified in CYP as a result of such testing.

A systematic review by Vlak *et al* ⁽¹⁾ estimated an overall prevalence of unruptured intracranial aneurysms (ICA) of 3.2% in a population without comorbidity. They calculated a prevalence ratio (PR) of 6.9 (95%CI 3.5-14) in patients with ADPKD compared to the population without comorbidity. Further analysis showed that the PR for patients with ADPKD and a family history of SAH or UIA was 2.0 (95%CI 0.5-7.4) compared with patients with ADPKD but no family history of SAH or UIA. The committee agreed that studies assessing the prevalence of SAH and ICA in CYP<18y with ADPKD should be undertaken prior to making recommendations in this area

References

1. Vlak M, Algra A, Brandenbur R et al. Prevalence of unruptured intracranial aneurysms, with emphasis on sex, age, comorbidity, country, and time period: a systematic review and meta-analysis. *Lancet Neurol*. 2011 (Jul); 10: 626-36.



7. Lay Summary

Autosomal Dominant Polycystic Kidney Disease (ADPKD) is thought to affect about 1 in 1,000 people in the UK. ADPKD causes a progressive decline in kidney function, with kidney failure tending to occur in middle age. Children and young people with ADPKD may not have any symptoms. However they may have high blood pressure, which may accelerate progression to later stages of chronic kidney disease.

There is uncertainty and variation in how health professionals manage children and young people with confirmed or a family history of ADPKD, because of a lack of evidence. For example, health professionals may be unsure about when to test children's blood pressure and how often to monitor it in the hospital clinic or at the GP. They may have different approaches in recommending scanning or genetic testing for ADPKD in childhood, with some recommending waiting until the young person is mature enough to make this decision his or herself.

This guideline is intended to help families affected by ADPKD by making sure that:

- health professionals with specialist knowledge in ADPKD offer you information on inheritance and potential benefits and harms of testing for ADPKD.
- the decision to test and the method of testing for ADPKD in children and young people is shared between you or your family and the health professionals
- blood pressure assessment is undertaken regularly in children and young people at risk of developing ADPKD